Author’s response to reviews

Title: A case of systemic lupus erythematosus initially presented with eosinophilic enteritis

Authors:

mehrnaz asadi (asadi_m@tums.ac.ir)
pejhman abdollahi (abdollahip@student.tums.ac.ir)
mohammad kalani (kalanim@tums.ac.ir)

Version: 5 Date: 16 September 2010

Author’s response to reviews:

1. I added the ethnicity of the patient and removed any irrelevant data from her demographic features for example her religion.

2. I explained more about the pathogenesis and presentation of lupus and removed the previous phrases that oversimplified SLE pathogenesis.

3. I added the titer of Antinuclear antibodies and the result of other extractable nuclear antigens.

4. The level of eosinophilia and detailed workup to detect the reason to justify increased eosinophil count and rule out the presence of parasitic infection were added.

5. Unfortunately, the photomicrograph of eosinophilic enteritis was not ready to be uploaded as the pathologist is on vacation and unavailable for a while, I will send it as soon as possible.

6. I added the result of kidney biopsy and therapeutic regimen administered to her

7. I improved to some extent the writing style

7. The changes are in blue color to distinguished from the old phrases.